CSE 322:

In tro duction to F ormal Mo dels in Computer Science

Spring 1998

Course Information

Instructor: Martin T ompa Sieg 426B T u 11:00{ 12: 00 tompa@cs.w ashington.edu 543- 926 3

T A: Jong Hee Kang Sieg 326D Th 2:30{ 3:3 0 jhk ang@cs.w ashington.edu

Prerequisites: CSE 321.

Grading: Homew ork (40%), Midterm (20%), Final (40%). P ercen tag es are appro ximate. The

�nal exam is sc heduled for T uesda y , June 9, 1998 , 2:30{4 :20 p.m.

Course W eb: All handouts and a log of all mail sen t to the class mailing list will also b e a v ai lable

on the course w eb at the follo wing URL:

http://www.cs.washington.edu/education/courses/322/CurrentQtr/

Mailing List: The class mailing list is cse322@cs.washington.edu. Instructions for subscribing

to the class mailing list can b e found on the course w eb.

W e will use this list for announcemen ts of general in terest to the class. Studen ts should also

feel free to use it to ask questions, p ost information, or initiate discussions of general in terest to

the class. Questions or commen ts that are not of general in terest should instead b e directed to the

T A or instructor at the addresses ab o v e.

T extb o ok: Harry R. Lewis and Christos H. P apadimitriou, Elements of the The ory of Compu-

tation, Second Edition, Pren tice-Hal l, 1998.

Reading assignmen t: By W ednesda y , read Sections 1.1 { 1.6. This should largely b e a review

of material from CSE 321.

If y ou w ould lik e to request academic accommo dations due to a disabilit y , please con tac t Disabled

Studen t Services, 448 Sc hmitz, 543{8 92 4 (V/TDD) . If y ou already ha v e a letter from Disabled

Studen t Services indicating y ou ha v e a disabilit y that requires academic accommo dati ons, please

let the instructor kno w so w e can discuss the appropriate accommo datio ns.